Author's response to reviews

Title: Plasma calprotectin and its association with cardiovascular disease manifestations, obesity and the metabolic syndrome in type 2 diabetes mellitus patients

Authors:

- Lise LP Pedersen (lise-pedersen@rsyd.dk)
- Mads MN Nybo (mads.nybo@rsyd.dk)
- Mikael MKP Kjæer Poulsen (mikael.kjaer.poulsen@rsyd.dk)
- Jan Erik JEH Henriksen (jan.erik.henriksen@rsyd.dk)
- Jordi JD Dahl (jordi.d Dahl@gmail.com)
- Lars LMR Melholt Rasmussen (lars.melholt.rasmussen@rsyd.dk)

Version: 4 Date: 11 December 2014

Author's response to reviews:

The full name and affiliation of the ethics comitee that approved the study have been added to the Methods section:

De videnskabsetiske Komitéer for Region Syddanmark